Author's response to reviews

Title: Self-Management in aged care settings: a design for a cluster randomised controlled trial of a psychosocial intervention to improve social participation and autonomy among seniors with a dual sensory impairment

Authors:

Lieve M. Roets-Merken (l.roets-merken@iq.umcn.nl)
Maud J.L. Graff (m.graff@iq.umcn.nl)
Sytse U. Zuidema (s.u.zuidema@umcg.nl)
Pieter G.J.M. Hermsen (p.hermsen7@chello.nl)
Steven Teerenstra (s.teerenstra@ebh.umcn.nl)
Gertrudis I.J.M. Kempen (g.kempen@maastrichtuniversity.nl)
Myrra J.F.J. Vernooij-Dassen (m.vernooij-dassen@iq.umcn.nl)

Version: 2 Date: 1 May 2013

Author's response to reviews:

The title page and abstract has been added to the manuscript.